Author's response to reviews

Title: Using multiple household food inventories to measure food availability in the home over 30 days: a pilot study

Authors:

Cheree Sisk (chereesisk@gmail.com)
Joseph R Sharkey (jrsharkey@srph.tamhsc.edu)
William A McIntosh (wamcintosh@ag.tamu.edu)
Jenna Anding (j-anding@tamu.edu)

Version: 4 Date: 8 April 2010

Author's response to reviews:

Changes made as requested to the title page